Author's response to reviews

Title: Evaluation of computerized health management information system for primary health care in rural India

Authors:

Anand Krishnan (kanandiyer@yahoo.com)
Baridalyne Nongkynrih (baridalyne@gmail.com)
Kapil Yadav (dr_kapilyadav@yahoo.co.in)
Satyavir Singh (satyavir@medialabasia.in)
Vivek Gupta (drvivek.gupta@gmail.com)

Version: 11 Date: 13 November 2010

Author's response to reviews:

13th November 2010

To
The Scientific Editor
BMC series journals

Subject: MS: 1250415418367419- Submission of revised manuscript

Dear Tim Shipley

Thank you for giving us the opportunity to revise the manuscript “Evaluation of Computerized Health Management Information System for Primary Health Care in rural India”.

We are hereby submitting the revised version (13th Nov 2010) after final proof reading

Thank you

Dr Baridalyne Nongkynrih